Author's response to reviews

Title: Use of the intraosseous screw for unilateral upper molar distalization and found well balanced occlusion: case report

Authors:

Ibrahim Erhan Gelgor (egelgor@hotmail.com)
Ali Ilhya Karaman (ihyaka@yahoo.com)
Tamer Buyukyilmaz (tbuyukyilmaz@superonline.com)

Version: 4 Date: 2 November 2006

Author's response to reviews:

The changes were made in the title as suggestions of the Editorial Production Team, and in the manuscript as suggestions of the Reviewer 1.